Author’s response to reviews

Title: TASKA: A modular task management system to support health research studies

Authors:

João Almeida (joao.rafael.almeida@ua.pt)
Rosa Gini (rosa.gini@ars.toscana.it)
Guiseppe Roberto (giuseppe.roberto@ars.toscana.it)
Peter Rijnbeek (p.rijnbeek@erasmusmc.nl)
José Luis Oliveira (jlo@ua.pt)

Version: 2 Date: 13 Jun 2019

Author’s response to reviews:

Editor Comments:

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

** Response: Completed. The funding bodies had no role in the design of the study and collection, the analysis, interpretation of the data and in writing the manuscript. We also added this information in the funding section of the manuscript.

-- We noticed that the corresponding author information provided on your title page is different to the one you indicated in the editorial submission. Please amend one so that they match.
** Response: Corrected. In this submission we placed the authors in the right order, according to the submitted manuscript.

-- Unfortunately the BMC Series does not accept footnotes for publication. Please remove the footnotes and include the information in a different format. Please note BMC Series does publish endnotes.

** Response: Corrected. We removed the footnotes since they are only the URL of the applications described in the manuscript, which can also be accessed through the respective reference.

-- At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different
colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

**Response: Please find the clean version in this submission. And many thanks for your care.**